Author’s response to reviews

Title: Functional constipation in infancy and early childhood: Epidemiology, Risk Factors, and Healthcare Consultation

Authors:

Anne Walter (hannahw91@hotmail.com)
Anne Hovenkamp (annehovenkamp@gmail.com)
Niranga Devanarayana (niranga1230@lycos.com)
Roshani Solanga (ragamamoh@yahoo.com)
Shaman Rajindrajith (shamanrajindrajith4@gmail.com)
Marc Benninga (m.a.benninga@amc.uva.nl)

Version: 2 Date: 30 Jul 2019

Author’s response to reviews:

Hereby, we provide the requested point-to-point response to each reviewer/editorial point raised.

The following comments were raised:

Comment 1:

-- Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Answer:

We have made the changes as requested by the editor. Authors’ contributions section, page 24, line 462-468.
Comment 2:

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Answer:

The suggested statement was included. Authors’ contributions section, page 24, 469.

Comment 3:

-- Thank you for providing a response to the reviewers/Cover Letter/Ethics Approval. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.

Answer:

Thank you for the attention. The document was removed from the submission files.